Author's response to reviews

Title: Does the Swedish Interactive Threshold Algorithm (SITA) accurately map and monitor visual field loss attributed to Vigabatrin.

Authors:

Miriam L Conway Dr. (miriam.conway.1@city.ac.uk)
Sarah L Hosking Prof. (shosking@vsg.org.au)
Robert P Cubbidge Dr. (r.p.cubbidge@aston.ac.uk)

Version: 2 Date: 19 November 2013

Author's response to reviews:

Dear Editor,

We have changed the following:

1) Title page included in manuscript
(2) Abstract included in manuscript
(3) Conclusion included in manuscript
(4) Competing interests included in manuscript
(5) Authors' Contribution included in manuscript
(6) Tables included in the manuscript.

Many thanks for your help,

Dr. Miriam Conway